Author's response to reviews

Title: Pubic Osteomyelitis due to Streptococcus Viridans after dental extraction; A novel observation

Authors:

Naseem Naqvi (naseemnaqvi@doctors.org.uk)
Rizwana Naqvi (drriz00@yahoo.com)
Christopher Wong (Christopher.Wong@aintree.nhs.uk)
Sushmita Pearce (Sushmita.Pearce@wwl.nhs.uk)

Version: 3 Date: 25 November 2007

Author's response to reviews:

Pubic osteomyelitis is a very rare osteomyelitis of bone. At risk groups include IV drug abusers, Diabetics and patients post Urological/Obstetrical procedures. Another, less well-known predisposing factor is strenuous physical activity in athletes. The pathogens are dependent on the predisposing conditions. The most common pathogen in athletes is Staphylococcus aureus. We describe the unique case of a patient with pubic osteomyelitis due to Streptococcus Viridans, which developed after dental extraction.